Author’s response to reviews

Title: Determinants of appropriate complementary feeding practice among mothers having children 6-23 months of age in rural Damot Sore district, Southern Ethiopia; a community based cross sectional study

Authors:

Dereje Teferi (yohannesdereje1979@gmail.com)

Abate Ashango (abate.areja11@gmail.com)

Mulugeta Kabalo (yohannismulugeta@yahoo.com)

Version: 1 Date: 24 Apr 2017

Author’s response to reviews:

Cover letter

Dear all editors and concerned bodies of BMC nutrition, we are resubmitting our paper to your journal because after editorial comment. There may be similarities with other researches especially the standard and operational definitions because it uses the same source and changing it may significantly affect the study. We will adhere to all rule and regulations of your journal and declare that the same article is not submitted to other journal. Furthermore, we declare that there is no competing of interest among authors and it’s our original work. So, let you consider it and process further based on your procedure.

Best regards

Corresponding author